Author's response to reviews

Title: Is hepatotropic contrast enhanced MR a more effective method in differential diagnosis of hemangioma than multi-phase CT and unenhanced MR?

Authors:

Edyta Szurowska (eszurowska@gumed.edu.pl)
Tomasz Nowicki (tnowicki@gumed.edu.pl)
Ewa Izycka-Swieszewska (eczis@wp.pl)
Joanna Wypych (jwypych@gumed.edu.pl)
Anna Drobinska-Jurowiecka (annad@gumed.edu.pl)
Karolina Markiet (karolina.markiet@gumed.edu.pl)
Arkadiusz Szarmach (a.szarmach@gumed.edu.pl)
Michal Studniarek (mstud@gumed.edu.pl)

Version: 4 Date: 28 March 2011

Author's response to reviews:

1. 'Conclusions' section was added after the Discussion.
2. All highlighting and tracking were removed from the manuscript.
3. 'Introduction' was chaned into 'Background'.
4. 'Material and methods' was chaned into 'Methods'.
5. 'Tab x' into 'Table x'.
6. 'Fig x' into 'Figure x'.
7. Figures’ order was corrected.